Author's response to reviews

Title: Emergency Department Triage: an Ethical Analysis

Authors:

Ramesh P Aacharya (raacharya@healthnet.org.np)
Chris Gastmans (Chris.Gastmans@med.kuleuven.be)
Yvonne Denier (Yvonne.Denier@med.kuleuven.be)

Version: 7 Date: 29 September 2011

Author's response to reviews:

Dear Sir/Madam,

I have done minor revisions as follows:
1) Abbreviations have been updated.
2) Font size of the letters in the references have been reduced to 10.
3) The gap page after the figure 1 has been removed.

Thank you very much for your kind attention and continued cooperation.

Sincerely,

Ramesh Prasad Aacharya